Author's response to reviews

Title: Integrative analysis of next generation sequencing for small non-coding RNAs and transcriptional regulation in Myelodysplastic Syndromes

Authors:

Dominik Beck (d.beck@student.unsw.edu.au)
Steve Ayers (sdayers@tmhs.org)
Jianguo Wen (jwen@tmhs.org)
Miriam B Brandl (miriam.brandl@student.unsw.edu.au)
Tuan D Pham (t.pham@adfa.edu.au)
Paul Webb (pwebb@tmhs.org)
Chung-Che Chang (jeffchang@tmhs.org)
Xiaobo Zhou (xzhou@tmhs.org)

Version: 5 Date: 17 February 2011

Author's response to reviews: see over
Dear Editor,

Thank you for the helpful suggestion. Please find below our point to point response to the requested revisions.

**Major revisions (we require the author to make these changes)**

1. **Author list:** the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Currently they do not match.

   The author names have been adjusted in the submission system and manuscript.

2. **Abstract:** this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusions.

   The abstract was changed accordingly.

3. **Abstract:** please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system. Currently they do not match.

   The modified abstract was updated in the submission system and the manuscript.

4. **Additional files:** all additional files must be mentioned in the text in numerical order, or removed from the system. There are 7 additional files uploaded into the submission system but only 3 additional files have been cited in the manuscript. Please ensure that all additional files are cited in consecutive numerical order in the manuscript or removed from the system.

   The additional files (including files containing the dataset produced for this study) have now been referenced accordingly in the document.